SPECIAL ISSUE ARTICLE

WILEY

Detecting and managing the patient with chronic kidney disease in primary care: A review of the latest guidelines

¹School of Cardiovascular and Metabolic Health, University of Glasgow, Glasgow, UK ²Glasgow Renal and Transplant Unit, NHS Greater Glasgow and Clyde, Glasgow, UK ³School of Health and Wellbeing, University of

Correspondence

Glasgow, Glasgow, UK

Jennifer S. Lees, School of Cardiovascular and Metabolic Health, University of Glasgow, 126 University Avenue, Glasgow, G12 8TA, UK. Email: jennifer.lees@glasgow.ac.uk

Kaitlin J. Mayne MBChB^{1,2} | Peter Hanlon PhD³ | Jennifer S. Lees PhD^{1,2}

Abstract

Chronic kidney disease (CKD) is a major global health problem, affecting about 9.5% of the population and 850 million people worldwide. In primary care, most CKD is caused by diabetes and/or hypertension, but a substantial proportion of cases may have alternative causes. During the early stages, CKD is asymptomatic, and many people are unaware that they are living with the disease. Despite the lack of symptoms, CKD is associated with elevated risks of cardiovascular disease, progressive kidney disease, kidney failure and premature mortality. Risk reduction strategies are effective and cost-effective but require early diagnosis through testing of the estimated glomerular filtration rate and albuminuria in high-risk populations. Once diagnosed, the treatment of CKD centres around lifestyle interventions, blood pressure and glycaemic control, and preventative treatments for cardiovascular disease and kidney disease progression. Most patients with CKD should be managed with statins, renin-angiotensin-aldosterone system inhibitors and sodiumglucose cotransporter-2 inhibitors. Additional treatment options to reduce cardiorenal risk are available in patients with diabetes, including glucagon-like peptide-1 receptor agonists and non-steroidal mineralocorticoid receptor antagonists. The Kidney Failure Risk Equation is a new tool that can support the identification of patients at high risk of progressive kidney disease and kidney failure and can be used to guide referrals to nephrology. This review summarizes the latest guidance relevant to managing adults with, or at risk of, CKD and provides practical advice for managing patients with CKD in primary care.

KEYWORDS

cardiovascular disease, glucagon-like peptide-1, lipid-lowering therapy, primary care, sodium glucose cotransporter-2 inhibitor, diabetic nephropathy

Plain Language Summary

The kidneys contain millions of tiny filters that remove waste products from our blood, making sure we have the right amount of water and nutrients in our bodies. Chronic kidney disease (CKD) is a condition where the kidneys do not work as well as they should over a long period of time. When a person has CKD, the kidneys do not filter waste products from the blood effectively, the kidneys can leak albumin (a type of protein) into the urine, and there can be a buildup of fluid and a loss of nutrients from the body. All these features mean that our bodies cannot keep as healthy as they should.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

© 2024 The Authors. Diabetes, Obesity and Metabolism published by John Wiley & Sons Ltd.

CKD is a very common problem. It is thought that about 1 in 10 people may be living with this disease. CKD is more likely to develop as we get older, particularly in people who have diabetes and high blood pressure. However, CKD can also occur alongside other medical problems, can develop unexpectedly because of uncommon conditions, or can be inherited from family members. Most people with CKD do not have any symptoms and may not know that they have the condition.

CKD can be identified by testing for waste products in the blood (to provide an estimate of the kidneys' filtering capacity) and testing for albumin in the urine (a marker of whether the filters have become damaged). These blood and urine tests can guide how severely a person has been affected by CKD. Detecting CKD is important because, even without symptoms, CKD can increase the likelihood of a person developing other medical problems, such as heart attacks and strokes. Although it is uncommon, about 1 in 100 people with CKD may also go on to develop 'kidney failure'. Kidney failure describes a situation where external support for kidney function, such as dialysis or a kidney transplant, may be needed.

If a health care provider detects that a person has CKD, treatment aims to protect the remaining kidney function for as long as possible and to reduce the risk of heart attacks and strokes. The health care provider will recommend living a healthy lifestyle, including eating a balanced diet, keeping physically active, maintaining a healthy body weight and avoiding smoking. People with CKD should have their blood pressure checked and should receive treatment if the blood pressure is too high for them. Cholesterol-lowering tablets are recommended for most patients. For some individuals, other effective medicines may be recommended, depending on that person's level of risk of developing other medical problems. Blood tests, urine tests and blood pressure should be monitored about once per year, or sometimes more often than this if the kidneys are more severely damaged.

Most patients with CKD will be cared for by their community health care provider, but some patients may need to be seen by a specialist kidney doctor. Risk calculators can help community health care providers decide which patients most need to be seen by a specialist.

Health care providers are advised to offer treatments that they think will benefit an individual, but they should also consider the person's wishes and expectations when deciding on a course of treatment.

1 | INTRODUCTION AND SCOPE OF REVIEW

1.1 | Global burden of chronic kidney disease

Globally, chronic kidney disease (CKD) is thought to affect about 9.5% of the population, or about 850 million people in total.¹ In the context of ageing populations and the increasing prevalence of multimorbidity (including among younger people), the prevalence of CKD is set to rise further. By 2040, CKD is expected to be the fifth leading cause of death globally.

1.2 | Why target chronic kidney disease?

CKD has a huge and growing economic impact. In the UK, the annual cost of CKD to the National Health Service is about £1.45 billion, through a combination of direct (screening, monitoring and managing complications) and indirect health care costs.² Although kidney failure is an uncommon consequence of CKD, dialysis treatment alone uses >50% of the budget spent on CKD. In the UK, CKD is responsible for approximately 45 000 premature deaths each year, with markedly higher rates of hospitalization and longer hospital stays among people with CKD.^{2,3} Disability and loss of productivity associated with the direct and indirect consequences of CKD lead to a loss of the working population and increased sick pay.

Not all cases of CKD or kidney failure are preventable; however, risk reduction strategies are effective and cost-effective if CKD is detected and managed in the early stages of the disease. Fewer than half of countries recognize CKD as a public health priority, and only a quarter have established CKD management strategies in place.¹ Appropriate and timely management is further limited by the burden of disease and widespread uncertainty about how to manage patients with CKD among non-specialists.

This article will summarize the latest guidance relevant to adults with, or at risk of, CKD. We will cover highlights of the major UK and international guidelines, with some discussion of the relevant points in the guidelines for comorbid conditions commonly seen in people with CKD (including hypertension, diabetes and heart failure). Through focused discussions with primary care health care practitioners in the UK, we have identified some of the major areas of uncertainty in managing CKD in the community. We aim to provide simple, practical and pragmatic advice to aid the non-specialist in managing patients with CKD (or at risk of CKD) in their day-to-day practice.

2 | IDENTIFICATION OF CHRONIC KIDNEY DISEASE

2.1 | Definition, classification and screening of chronic kidney disease

CKD is defined as an abnormality in kidney structure or function, that is present for a minimum of 3 months and has implications for health.⁴ International guidelines recommend consideration of cause as part of the CKD definition and determination of management (see Section 3.1 below). In most cases of CKD managed in primary care, the implications for health are determined by the values of two measures: the estimated glomerular filtration rate (eGFR; as a marker of kidney filtration) and albuminuria (as a marker of glomerular damage). As eGFR declines and/or as albuminuria increases, the risk of all important adverse outcomes associated with CKD rises. This effect is multiplicative, such that people with both low eGFR and high albuminuria are at the highest level of risk. A recent meta-analysis of over 27 million individuals in 114 global cohorts confirmed risks associated with CKD (by eGFR and/or albuminuria criteria) extended to acute kidney injury (AKI), kidney failure replacement therapy, atrial fibrillation, peripheral artery disease, myocardial infarction, stroke, heart failure, cardiovascular mortality, all-cause hospitalization and all-cause mortality.⁵

A threshold of clinically significant CKD is set at eGFR <60 ml/ min/1.73 m² and/or albuminuria >3 mg/mmol,^{4,6} as these are the thresholds below and above which the risk of adverse events increases, respectively. A patient can therefore have clinically important CKD even if the eGFR is \geq 60 ml/min/1.73 m². A reminder of the classification of CKD, including nomenclature, is shown in Figure 1.

CKD is usually asymptomatic until later stages of the disease; symptoms would not be expected unless eGFR was less than about 25 ml/min/1.73 m²; albuminuria is usually entirely asymptomatic except in the nephrotic syndrome (see Section 2.3 below). For this reason, screening for CKD in high-risk populations is recommended: patients with type 1 or type 2 diabetes, hypertension, previous AKI, cardiovascular disease, structural urinary tract disease, gout, multisystem diseases with potential kidney involvement (e.g. systemic lupus erythematosus), family history of kidney failure or a hereditary kidney disease, or people with previous incidental findings of haematuria or proteinuria.⁶ The guidelines are not specific about the frequency of screening in each of these populations, but it would be reasonable to offer screening with eGFR and albuminuria testing annually in these groups. CKD screening with eGFR and albuminuria testing has been shown to be a cost-effective strategy, at least in people with diabetes and hypertension.⁷ eGFR testing should also be offered on an annual basis to adults taking treatment that could affect kidney function [calcineurin inhibitors, lithium, or non-steroidal anti-inflammatory drugs (NSAIDs)].⁶

Further discussion of eGFR and albuminuria to detect and classify CKD, including interpretation and guidance for repeat testing, is detailed in the sections below.

2.2 | Estimated glomerular filtration rate

Creatinine is a cheap and widely available filtration marker and creatinine-based eGFR (eGFRcr) is recommended for first-line testing and routine monitoring of CKD in primary care.^{4,6} Over the last two decades, various groups have developed and validated several equations to calculate eGFRcr. In the UK, the equation recommended for use is the Chronic Kidney Disease Epidemiology Collaboration (CKD-EPI) equation 2009,⁸ although international guidelines recommend the updated CKD-EPI equation 2021 without inclusion of a race coefficient.⁹ In practice, the value reported to primary care practitioners will usually be dictated by the local laboratory conducting the test.

		Pers		stent albuminuria		
				A1	A2	A3
				Normal-mildly ↑	Moderately ↑	Severely ↑
				<3 mg/mmol <30 mg/g	3-30 mg/mmol 30-300 mg/g	>30 mg/mmol >300 mg/g
te	G1	Normal or high	≥90			
Glomerular filtration rate (ml/min/1.73m ²)	G2	Mildly ↓	60-89			
	G3a	Mildly-moderately \downarrow	45-59			
	G3b	Moderately-severely \downarrow	30-44			
	G4	Severely ↓	15-29			
0	G5	Kidney failure	<15			

FIGURE 1 Classification of chronic kidney disease (CKD) according to glomerular filtration rate and albuminuria categories. This heatmap contains illustrative information about relative risks of adverse kidney, cardiovascular and mortality outcomes compared with a person without CKD. This heatmap does not provide information about the absolute risks, or competing risks, of these events: absolute and competing risks may also vary by other factors including age, sex, burden of comorbidity and/or frailty. Green, low risk (if no other markers of kidney disease, no CKD); yellow, moderately increased risk; orange, high risk; red, very high risk.

On a population level, eGFRcr estimates kidney function reasonably accurately⁸⁻¹¹; however, it is worth being aware of some issues with creatinine as a marker of kidney function to guide interpretation. Creatinine is subject to significant (at least 5%) within-individual biological and analytical variability. Measurement inconsistencies of both creatinine and eGFRcr is therefore expected and should be acknowledged in the interpretation of repeat samples.⁶

There are various situations when eGFRcr is probably less accurate. In settings where creatinine is reduced (e.g. people with muscle wasting disorders, including cachexia, neurological disease and sarcopenia seen with ageing, people with limb amputations, malnourished adults), eGFRcr will be higher, suggesting better kidney function than is the case in reality. In individuals with elevated creatinine (e.g. high muscle mass in athletes or those who use protein supplements), eGFRcr will be lower, suggesting poorer kidney function than is true. Creatinine can also become elevated, with apparently reduced eGFRcr, in patients receiving drugs that block tubular creatinine secretion ('pseudo-AKI'), but in the absence of genuine kidney damage. Trimethoprim is the most common culprit in primary care, but this effect can also be seen with some treatments for the human immunodeficiency virus and tyrosine kinase inhibitors.¹²

In these scenarios, creatinine and eGFRcr results should be interpreted with caution.⁶ Inaccurate eGFRcr is associated with a higher likelihood of adverse outcomes,¹³ including medication-associated adverse events such as hyperkalaemia with co-trimoxazole treatment, the toxic effects of baclofen and high digoxin levels.¹⁴

There are no specific recommendations to use alternatives to creatinine for the estimation of kidney function in primary care in the UK. In other settings where highly accurate estimation of kidney function is required (e.g. living kidney donation, dosing of anti-cancer therapies), UK and international guidelines recommend consideration of measured GFR using an isotope clearance technique.^{4,6} Internationally, cystatin C testing is additionally recommended, including in primary care, when eGFRcr may be inaccurate, and is used in a combined equation with creatinine.⁴ Although cystatin C testing is available in the UK and of potential value in various clinical scenarios,¹⁵ a lack of evidence for its accuracy and cost-effectiveness in a UK population means that cystatin C testing is not routinely recommended in current National Institute for Health and Care Excellence (NICE) guidelines.⁶

2.2.1 | What to do if the estimated glomerular filtration rate is 'abnormal'?

Any kidney function test result should be interpreted in the context of previous results (where available) as well as the clinical situation.

If eGFR is <60 ml/min/1.73 \mbox{m}^2 and no previous results are available,

 If unwell, consider admission to hospital for further investigation and management.

- 2. If not unwell, re-test within 2 weeks.⁶
 - a. Is this AKI? Use clinical judgement in the interpretation of a changing kidney function test over 2 weeks. Review potential causes, e.g. recent medications [trimethoprim, NSAIDs, reninangiotensin-aldosterone system inhibitor (RAASi) or sodium glucose cotransporter-2 inhibitor (SGLT2i)]. Depending on the severity of AKI, hospital admission and/or specialist advice may be required.
- If blood test is stable at 2 weeks, re-test in 3 months (or ideally a further two times over a minimum of 3 months to determine the rate of progression).^{4,6}
 - a. Is this CKD? Check for albuminuria, and use a urine dipstick for non-visible haematuria (see Section 2.3 below). If CKD is confirmed on blood and/or urine tests, ensure CKD is coded in the clinical record (see Section 2.1 above) and manage as CKD (see Section 3 below).

2.3 | Albuminuria and non-visible haematuria

Albuminuria is both a marker of glomerular damage and a driver of further glomerular damage. The presence of albuminuria at any level is abnormal. With rising albuminuria, the risk of adverse outcomes (including cardiovascular disease, cancer, AKI, kidney disease progression and kidney failure, and all-cause mortality) increases, even when eGFR is normal.^{5,16,17} Small changes in albuminuria have a much more pronounced effect on these future outcomes than mild reductions in eGFR. Importantly, albuminuria is modifiable with treatment, and improvement in albuminuria over time is associated with a lower risk of a plethora of adverse outcomes.¹⁸ Some of the beneficial effects of treatments recommended for CKD are mediated, at least in part, by improvements in albuminuria.^{19,20} Improved detection, monitoring and treatment of albuminuria is therefore paramount in altering the disease course, morbidity and mortality associated with CKD.

Although albuminuria testing is clearly recommended in people with CKD and at high risk of CKD, the rate of albuminuria testing is suboptimal, and there are currently no evidence-based guidelines to recommend the frequency of albuminuria testing. Consensus reports suggest annual testing among people with diabetes,^{4,21} and this is achieved in about 70-80% of individuals with diabetes.²²⁻²⁴ However, guidance is less clear about the frequency of albuminuria testing required among people with hypertension without diabetes or CKD,^{6,25-27} and testing of albuminuria in this group is much lower (20-30%).^{22,23} As the prevalence of albuminuria in hypertension is similar to that seen in diabetes, annual albuminuria testing in people with hypertension would be a reasonable approach.

Urine albumin-creatinine ratio (UACR) is the preferred measure to detect glomerular disease. At low levels of albuminuria, the urine protein-creatinine ratio (UPCR; measuring proteinuria, including albuminuria and other urinary proteins) is insensitive. When UACR is ≥70 mg/mmol, UPCR can be used as an alternative.⁶ If UACR is not available, conversion calculators can approximate UACR from UPCR

MAYNE ET AL.

or urine dipstick protein (https://ckdpcrisk.org/pcr2acr/).²⁸ Urine dipstick should be performed as well as UACR to test for non-visible haematuria.4,6

2.3.1 What to do if you detect albuminuria?

- 1. Confirm albuminuria on an early morning urine sample.
- 2. Consider the cause: is there pre-existing diabetes or hypertension? Does the patient have a cause for visible haematuria, such as menstruation? Could they have a urinary tract infection? Have they participated in vigorous exercise?⁴
- 3. Where the presence of albuminuria is confirmed (UACR ≥3 mg/mmol).
 - a. Manage blood pressure to target (see Table 1), starting with renRAASi.
 - b. SGLT2i after introduction of RAASi (if eGFR ≥20 ml/ $min/1.73 m^2$)
 - c. The frequency of testing should be guided by severity (see Table 2) and the potential for improvement with blood pressure control. Although not specifically advised in official guidance, it is appropriate to repeat UACR testing while uptitrating antihypertensive treatment. This will inform both the intensity of the required blood pressure control and the likelihood of a requirement for referral to a nephrologist after the blood pressure target has been achieved.

- 4. If UACR ≥30 mg/mmol (UPCR >50 mg/mmol) with non-visible haematuria, refer to nephrology.^{4,6}
- 5. If UACR ≥70 mg/mmol (UPCR ≥100 mg/mmol) after blood pressure control,^{4,6} including if UACR is in the nephrotic range but there is no evidence of nephrotic syndrome, refer to nephrology. One caveat is that referral may not be required if the elevated UACR is known to be caused by diabetes and the patient is already on appropriate treatment with RAASi +/- SGLT2i.⁶
- 6. If there is new evidence of nephrotic syndrome (spot UACR >200 mg/mmol or UPCR >300 mg/mmol, hypoalbuminaemia <30 g/L and peripheral oedema), this warrants urgent referral to nephrology. Depending on the centre, an assessment will often be offered within days for further investigation +/- renal biopsy.

2.3.2 Non-visible haematuria without proteinuria

Where non-visible haematuria $(\geq 1+)$ is identified on dipstick in the absence of proteinuria, this is 'persistent' when the dipstick is positive in two of three tests.⁶ Persistent non-visible haematuria in requires urgent further investigation on suspicion of urinary tract malignancy in individuals >60 years.³² Otherwise, persistent non-visible haematuria should be monitored annually with repeat dipstick, eGFR, albuminuria or proteinuria testing, and blood pressure monitoring, unless the non-visible haematuria resolves.⁶

TABLE 1 Blood pressure targets from major guidelines in chronic kidney disease and hypertension.

Treatment targets	KDIGO ^{4,29}	ACC/AHA ³⁰	ESC/ESH ^{27,31}	ISH ²⁶	NICE ²⁵
General population	NA	<130/80	<130/80	<130/80	<140/90
					<130/80 if UACR ≥70 mg/mmol
CKD	<120 systolic ^a	<130/80	<130/80	<130/80	<140/90
					<130/80 if UACR ≥70 mg/mmol

Note: All guidelines generally recommend considering less intensive individualized targets in patients with advanced age and/or frailty, high risk of falls and fractures, very limited life expectancy or symptomatic postural hypotension.

Abbreviations: ACC/AHA, American College of Cardiology/American Heart Association; CKD, chronic kidney disease; ESC/ESH, European Society of Cardiology/European Society of Hypertension; ISH, International Society of Hypertension; KDIGO, Kidney Disease: Improving Global Outcomes; NA, not applicable; NICE, National Institute for Health and Care Excellence; UACR, urine albumin-creatinine ratio. ^aIn patients with functioning kidney transplants, the target should be <130 systolic and <80 diastolic.

TABLE 2 Suggested frequency of monitoring of chronic kidney disease.

	UACR (mg/mmol)				
eGFR, ml/min/1.73 m ²	ACR category A1 <3	ACR category A2 3-30	ACR category A3 >30		
eGFR category G1 ≥90	Annual	Annual	4-monthly (\times 3 per year)		
eGFR category G2 60-89	Annual	Annual	4-monthly ($ imes$ 3 per year)		
eGFR category G3a 45-59	Annual	6-monthly ($ imes$ 2 per year)	4-monthly (\times 3 per year)		
eGFR category G3b 30-44	6-monthly ($ imes$ 2 per year)	4-monthly ($ imes$ 3 per year)	4-monthly (\times 3 per year)		
eGFR category G4 15-29	4-monthly ($ imes$ 3 per year)	4-monthly ($ imes$ 3 per year)	At least 3-monthly (≥4× per year)		
eGFR category G5 <15	At least 3-monthly (≥4× per year)	At least 3-monthly (≥4× per year)	At least 3-monthly (≥4× per year)		

WII FY

4631326, 0, Downloaded from https://dom-pubs.onlinelibrary.wiley.com/doi/10.1111/dom.15625 by Ministry Of Health, Wiley Online Library on [15/05/2024]. See the Terms

and Conditions

i (https:

library.wiley.com

and

-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

⁶ ____WILEY-

2.4 | Coding

Coded CKD is associated with better quality CKD and cardiovascular disease management: better blood pressure control, higher rates of UACR testing, prescribing of RAASi and statins, lower prescribing of NSAIDs and lower rates of other medication and dosing errors.^{22,33,34} However, recent evidence shows that biochemical CKD is not coded in the clinical record of 35-55% of individuals^{22,33} and is less probable in women compared with men, younger people, people from areas of greater socio-economic deprivation, and non-White racial/ethnic groups.^{33,35} Underlying the importance of risk recognition, coding of CKD in the clinical record is recommended⁴ and can be improved through quality improvement interventions.³⁶

2.5 | Recall and monitoring

There is little to no evidence to support clear guidelines for the frequency of monitoring in CKD. UK guidelines recommend a minimum frequency for monitoring according to the severity of eGFR and albuminuria abnormalities (Table 2); however, more frequent monitoring may be required in those with a faster rate of deterioration in kidney function, other risk factors (such as diabetes, heart failure and hypertension), and those with recent changes to their treatment (e.g. RAASi, NSAIDs, or diuretics).⁶ There is no evidence to guide the frequency of UACR testing, but this could be performed alongside eGFR monitoring or more frequently if the result may warrant titration of risk reduction strategies (see Section 3 below). Guidelines allow for some tailoring in monitoring and can consider patient preferences, including conservative management of advanced CKD.

3 | MANAGEMENT IN PRIMARY CARE

3.1 | Consider the cause

Most CKD seen in primary care in the UK is caused by diabetes and/or hypertension; however, it is important to be aware of the alternatives and avoid attributing the cause to diabetes and/or hypertension if the presentation is atypical (Figure 2). In patients with a new diagnosis of CKD, a full clinical history and examination, including social, environmental and family history, is recommended.⁴ Consider an obstructive cause if the patient has urinary tract symptoms. Rare or genetic causes of CKD make up about 5-10% of CKD seen in primary care but cause 25% of cases of incident kidney failure.³⁷ Consider the possibility of renal involvement in patients with relevant multisystem disorders (e.g. systemic lupus erythematosus). In situations where the cause is uncertain or an uncommon cause is suspected, refer to nephrology.

3.2 | When to offer a renal ultrasound scan in primary care

- Suspected structural abnormalities in the urinary tract, including urinary tract obstruction, should be considered when the patient has a history of urinary symptoms, including visible or persistent non-visible haematuria.^{4,6}
- Family history of polycystic kidney disease and over the age of 20 years (ultrasound is insensitive for the diagnosis of polycystic kidney disease in patients younger than this).⁶
- Presentation with advanced CKD (stage G4-5; eGFR <30 ml/ min/1.73 m²).⁶
- 4. Rapid progression of CKD (see Table 3).⁶

3.3 | Communicating diagnosis

Fewer than half of patients with CKD are aware that they have a problem with their kidneys, even among those at high risk of progressing to kidney failure.^{38,39} To promote better awareness and patient self-management, UK and international guidelines recommend educating patients about a diagnosis of CKD^{4,6} and the implications for cardiovascular disease risk. CKD will lead to kidney failure in only a minority of patients, and specific discussion about kidney replacement therapy (dialysis and kidney transplant) should be reserved for patients with severe or progressive CKD who require specialist care. There are some excellent patient resources providing CKD education, which can be accessed through professional organizations such as Kidney Care UK in booklets, online written and short video formats.⁴⁰

3.4 | Lifestyle

Following healthy lifestyle measures is the cornerstone of the management of all chronic diseases, including CKD. The evidence and guidance for lifestyle management in CKD have not changed substantially in recent years, and we will not perform a detailed review of these points here. In brief, major guidelines in CKD,^{4,6} diabetes²¹ and hypertension²⁵ recommend similarly that patients living with these conditions should follow healthy dietary patterns, engage in recommended levels of physical activity (including both aerobic and resistance training), avoid smoking and reduce body weight to avoid obesity.

3.5 | Cardiorenal risk reduction

CKD management has progressed enormously in recent years, with a selection of new therapeutics available to reduce the risk of CKD progression and prevent major adverse cardiovascular and renal events and reduce premature mortality. Irrespective of the underlying cause,



FIGURE 2 Classification of chronic kidney disease (CKD) according to cause.

1	uisease.	
	Starting eGFR, ml/min/1.73 m ²	Sustained ^a change in eGFR (ml/min/1.73 m ²) within 12 mont reflecting rapid progression ^b
	90	75

TABLE 3	Definition of rapid progression of chronic kidney
disease.	

Starting eGFR, ml/min/1.73 m ²	(ml/min/1.73 m ²) within 12 months reflecting rapid progression ^b
90	75
80	65
70	55
60	45
50	38
40	30
30	23
20	15

Abbreviation: eGFR, estimated glomerular filtration rate. ^aPersistent over at least 3 months, i.e. not reflecting acute kidney injury.

^bHigher value of: 15 ml/min/1.73 m² decline; 25% decline in eGFR and change in eGFR category.

blood pressure control with RAASi [angiotensin converting enzyme inhibitors (ACEi) or angiotensin receptor blockers (ARB), but not both], treatment with high-intensity statins and SGLT2i are indicated in almost all patients with non-dialysis CKD, with some additional treatment options in patients with type 2 diabetes, heart failure and/or in other special circumstances. We have included a simple illustration of treatments to consider among patients with CKD (Figure 3), with a more detailed discussion of the guidance in the subsections below.

3.5.1 Blood pressure

Depending on the stage and cause of CKD, hypertension in CKD is observed in 60-90% of individuals. As in the general population, hypertension in CKD should be diagnosed using standardized techniques, preferably using home or ambulatory readings.²⁵ Among people with CKD, blood pressure should be checked at least annually,



Atorvastatin 20 mg once daily

SGLT2 inhibitor

f eGFR >20

FIGURE 3 Management of CKD in primary care. CCB, calcium channel blocker; CKD, chronic kidney disease; CVD, cardiovascular disease; eGFR: estimated glomerular filtration rate in ml/min/1.73 m²; GLP-1RA, glucagon-like peptide-1 receptor agonist; RAASi, renin-angiotensin-aldosterone system inhibitor; SBP, systolic blood pressure; SGLT2i, sodium glucose cotransporter-2 inhibitor.

Finerenone

if type 2 diabetes, eGFR >25, albuminuria & normal

potassium

even if within the target range.^{4,6} The frequency of blood pressure monitoring should be greater where blood pressure is suboptimally controlled and where albuminuria is elevated, as these are modifiable risk factors for progressive kidney disease and other adverse outcomes.

In addition to RAAS inhibition:

High-intensity statin

almost All CKD

Additional therapies to consider in diabetes

RAASi remain the preferred choice, where tolerated, for management of hypertension in CKD, but particularly among those with detectable albuminuria, type 2 diabetes and/or heart failure. Blood pressure management for patients with CKD should otherwise follow standard guidance for the management of hypertension.²⁵⁻²⁷ For patients who have persistent uncontrolled or resistant hypertension (blood pressure above target on ≥4 agents, with or without CKD), a specialist referral is recommended for further assessment.⁶

Blood pressure targets have been subject to some controversy, and the recommended target varies across guidelines (Table 1). On average, the guidelines support a blood pressure target of <130/80 among people with CKD, with or without diabetes. Lower targets than this are generally not recommended in clinical practice because of an increased risk of adverse events. All guidelines recommend considering less intensive, individualized targets in patients with advanced age and/or frailty, a high risk of falls and fractures, a very limited life expectancy or symptomatic postural hypotension.

3.5.2 | Statins

8

Slowing CKD progression and reducing CVD risk

Currently, cardiovascular risk scores do not adequately consider CKD as a risk factor for cardiovascular disease; cardiovascular risk scores therefore do not change risk-reduction guidance in CKD.^{6,41,42} Current guidelines pragmatically suggest that people with CKD (eGFR <60 ml/min/1.73 m² and/or albuminuria) are considered high risk for cardiovascular disease. UK guidelines recommend atorvastatin 20 mg

once daily for primary prevention of cardiovascular disease among all people with CKD.⁴¹ In Kidney Disease: Improving Global Outcomes (KDIGO) guidelines, recommendations vary somewhat by age and CKD status: in people age ≥50 years, statin-based regiments are recommended for those with eGFR <60 and/or albuminuria: specifically, atorvastatin 20 mg, rosuvastatin 10 mg or simvastatin 20 mg/ ezetimibe 10 mg combination. In a meta-analysis of 28 trials of statin-based therapy, statin-based treatments showed no evidence of reducing major vascular events among patients treated with dialysis.⁴³ Statin-based therapies are not recommended for the primary prevention of cardiovascular disease in those treated with chronic dialysis⁴ but can still be prescribed safely for secondary prevention in those who have had a previous cardiovascular or cerebrovascular event.

GLP-1 RA

if type 2 diabetes requiring additional glycaemic control

3.5.3 | Renin-angiotensin-aldosterone system inhibitors

ACEi and ARB have beneficial effects in reducing proteinuria and are recommended in patients with moderate or severe albuminuria, including in the absence of diabetes or hypertension.^{4,6} In a metaanalysis of 119 trials (64 768 participants) in patients with CKD, ACEi and ARB reduced the odds of kidney failure, cardiovascular events and all-cause mortality.⁴⁴ ACEi were possibly superior to ARBs for these outcomes, although guidelines do not recommend one drug class over the other.^{4,6}

Optimization of RAASi in patients with CKD is often limited by two concerns, i.e. (a) decline in kidney function, and (b) hyperkalaemia. Kidney function should be checked within about 2 weeks of starting or dose-escalating RAASi,^{4,6} accepting a rise in serum creatinine of up to 30% or a decrease in eGFR of up to 25%, to maintain the prescribed dose of RAASi.⁶

CKD monitoring

Disease prevention

e.g. vaccination

UK guidelines recommend avoiding RAASi initiation if serum potassium is >5 mmol/L; however, international guidelines do not set a threshold.⁴ Dietary and pharmacological measures (including cation exchange compounds such as sodium zirconium cyclosilicate and patiromer) are recommended to reduce serum potassium to facilitate initiation and avoid dose reduction because of hyperkalaemia in patients who qualify for RAASi treatment.^{4,6} Because of the higher rates of adverse events, including AKI, hypotension and hyperkalaemia, RAASi (ACEi, ARB and direct renin inhibitors) should not be prescribed in combination with each other.⁶

3.5.4 | Sodium-glucose cotransporter-2 inhibitors

SGLT2is have been shown in populations of patients with CKD with and without diabetes to reduce kidney disease progression, AKI, cardiovascular risk, hospitalization with heart failure and all-cause mortality, irrespective of the primary kidney disease.^{42,45–51} In the UK (broadly matching international guidelines⁴), SGLT2i are now recommended for use in people with CKD (eGFR \geq 20 ml/min/1.73 m², excluding polycystic kidney disease, type 1 diabetes or patients with kidney transplants) in any of the following commonly overlapping clinical scenarios:⁴⁶ (a) type 2 diabetes; (b) symptomatic heart failure; (c) established coronary disease; (d) albuminuria (UACR \geq 25 mg/mmol or UPCR \geq 35 mg/mmol); and (e) eGFR 20-45 ml/min/1.73 m² in the absence of type 2 diabetes, heart failure or albuminuria.

Prescribing considerations

The guidelines are agnostic to the choice of SGLT2i. Treatment can be continued as eGFR declines until there is a requirement for dialysis or a kidney transplant.⁴⁶ The glycaemia-lowering effect of SGLT2i diminishes with declining eGFR; eGFR is approximately <30 ml/min/1.73 m², SGLT2is have no noticeable effect on HBA1c.⁵² Additional glucose-lowering therapy may be required in people with type 2 diabetes and CKD. In patients with eGFR >45 ml/min/1.73 m² and HbA1c <58 mmol/L, 50% dose reductions in sulphonylureas or meglitinides, and 20% reductions in insulin treatment, are recommended to avoid precipitating hypoglycaemia.⁴⁶ No dose adjustments are required when starting SGLT2i treatment in patients taking only metformin, pioglitazone, DPP-4/gliptins, or glucagon-like peptide-1 receptor agonist (GLP-1RA) therapy.⁴⁶ In patients without diabetes, it is not necessary to check or monitor HbA1c in response to treatment with SGLT2i.

Blood test monitoring is not required

Initiation of SGLT2i commonly causes a small initial reversible reduction in eGFR ('eGFR dip'), followed by stabilization of eGFR and improved kidney outcomes. Blood test monitoring is neither required nor recommended after initiation of SGLT2i, and treatment should not be discontinued if the eGFR dip has occurred after initiation of SGLT2i in the absence of other adverse features.

Sick day rules

Following sick day guidance may substantially reduce the potential risk of ketoacidosis.⁴⁶ Patients should be advised to withhold SGLT2i

if they are unwell (vomiting, diarrhoea, fever, sweats, shaking or not eating/drinking normally) or have planned restrictions on food intake (e.g. prolonged periods of fasting); however, they should be advised to restart treatment after the illness has subsided.

3.6 | Additional recommendations for patients with type 2 diabetes

3.6.1 | Non-steroidal mineralocorticoid receptor antagonists

The non-steroidal mineralocorticoid receptor antagonist (nsMRA) finerenone has been shown in two major trials in diabetic kidney disease to reduce the risk of major adverse kidney events (including kidney failure, a 40% reduction in eGFR and death from renal events) and cardiovascular events (including myocardial infarction, stroke, hospitalization from heart failure, and death from cardiovascular causes).^{53,54} Finerenone is newly recommended for use as an add-on to optimized standard care (including RAASi and SGLT2i) in patients with type 2 diabetes, eGFR >25 ml/min/1.73 m², albuminuria >3 mg/ mmol and normal serum potassium.^{4,55,56}

Prescribing considerations for nsMRA

In the UK, it is expected that finerenone treatment will be initiated in secondary care to begin with but will eventually be prescribed in primary care.⁵⁶ nsMRA are associated with lower rates of hyperkalaemia than older steroidal MRA (e.g. spironolactone and eplerenone), but potassium monitoring is recommended within 1 month of initiation and then every 4 months for the duration of treatment.⁴ If potassium rises to above 5.5 mmol/L, treatment should be suspended until potassium-lowering measures are implemented (e.g. dietary changes or potassium cation-exchange resins). Re-initiation of treatment can be considered when potassium is \leq 5 mmol/L. nsMRA can be continued at eGFR <25 ml/min/1.73 m², but with ongoing potassium monitoring and consideration of dose reduction if required. nsMRA should be stopped if eGFR falls below 15 ml/min/1.73 m² or if the patient requires dialysis or a kidney transplant.⁵⁶

3.6.2 | Glucagon-like peptide-1 receptor agonists

GLP-1RAs have proven to reduce major adverse cardiovascular events (including myocardial infarction, stroke and cardiovascular death), and there is some evidence of benefit to kidney outcomes (including kidney disease progression, kidney failure and death from kidney disease) among patients with type 2 diabetes and CKD.⁵⁵ GLP-1RAs are recommended for the treatment of patients with type 2 diabetes and CKD who are not meeting individualized HbA1c targets, and who are either already treated with or unable to tolerate metformin and SGLT2i.⁵⁵

Prescribing considerations for GLP-1RA

Gastrointestinal side effects, particularly nausea and diarrhoea, are very common. Guidelines recommend starting with a low dose of

GLP-1RA and titrating slowly ('start low, go slow') to reduce the likelihood of experiencing side effects.⁵⁵ Because of their popular, supporting role in aiding weight loss in patients living with overweight and obesity, GLP-1RAs are currently in very short supply.⁵⁷

3.6.3 | Dosing and prescribing

 \perp Wilfy_

Dosing

People with CKD may be at greater risk of the nephrotoxic effects of certain medications, and dose adjustment or drug avoidance may be necessary. The Renal Drug Database (https://www.renaldrugdatabase.com/s/) is a UK-based digital tool to aid prescribers in understanding prescribing and dosing considerations for patients with CKD and/or kidney failure. The resource was written by clinical renal pharmacists, supported by the UK Renal Pharmacy Group, and provides detailed advice and monographs for drug prescribing and de-prescribing in CKD.

Non-steroidal anti-inflammatory drugs

Clinical guidelines recommend the avoidance of prolonged use of NSAIDs among people with CKD stage G3 (eGFR \geq 30 ml/min/1.73 m²) and total avoidance in people with CKD stages G4-5 (eGFR <30 ml/min/1.73 m²). However, data confirming the nephrotoxicity of NSAIDs are limited, and alternative analgesic use (e.g. opioids) in people with CKD have their own risks. Judicious use of NSAIDs can be considered even in people with CKD stages G3-4, on a case-by-case basis and after careful consideration of the individual risk/benefit profile.⁵⁸ In CKD stage G5, NSAIDs should be avoided completely, except when providing palliative care.⁵⁸

Sick day rules

Sick day rules for SGLT2i are covered in the relevant section above. Despite a paucity of data for effectiveness in preventing AKI or other relevant outcomes, sick day rules are often recommended for sulphonylureas, RAASis, diuretics, metformin and NSAIDs. Sick day advice is often poorly understood, associated with medication errors, and particularly with failure to restart the medications. Patients should be clearly advised of the plan to recommence medication after the illness has subsided.⁴ Medication review after illness is recommended to avoid accidental discontinuation of treatments with important clinical benefits⁴; practically speaking, this may be best accomplished opportunistically while undertaking medical reviews for other reasons.

3.6.4 | Vaccinations

Patients with CKD, including patients on haemodialysis should be offered vaccination against influenza (CKD stages G3-5) and pneumococcus (CKD stages G4-5) every year.⁵⁹ Patients with CKD stages G4-5 should additionally be offered the hepatitis B vaccination.⁵⁹ In practice, this is usually reserved for patients who would probably proceed with kidney replacement therapy (dialysis or a kidney transplant) and will often be guided by nephrology services. The COVID-19 pandemic confirmed patients with kidney failure (requiring dialysis or a kidney transplant) to be vulnerable to severe disease, high mortality rates and an impaired response to vaccination compared with the general population.^{60,61} Patients with non-dialysis CKD also experienced higher rates of adverse outcomes, including severe AKI requiring dialysis treatment.⁶² Patients with CKD should be encouraged to receive additional COVID-19 vaccinations if/when they become available. Patients with CKD stage G4-5 are eligible for consideration of treatment for COVID-19 if confirmed positive.

3.7 | REFERRAL TO NEPHROLOGY

The referral criteria capture patients who require, or are suspected to require, specialist investigation, management or counselling relating to kidney disease. In the UK, these are as follows.⁶

- 5-year risk of needing kidney replacement therapy of >5%, estimated using the four-variable Kidney Failure Risk Equation (KFRE).
- UACR of ≥70 mg/mmol, unless known to be caused by diabetes and already appropriately treated.
- UACR of >30 mg/mmol, together with haematuria.
- Rapid progression of CKD (see Table 3):
- a sustained decrease in eGFR of ≥25% and a change in the eGFR category within 12 months
- $\circ~$ a sustained decrease in eGFR of ≥15 ml/min/1.73 m² per year
- hypertension that remains above the person's individual target despite the use of at least four antihypertensive medicines at therapeutic doses
- known or suspected rare or genetic causes of CKD
- suspected renal artery stenosis.

The main recent changes in the guidelines have been (a) the removal of the criterion to refer patients with eGFR <30 ml/min/1.73 m², and (b) the addition of KFRE as a marker of the likelihood of kidney disease progression and kidney failure in people with CKD.^{4,6} The KFRE is validated for use in people with eGFR <60 ml/min/1.73 m², using age, sex, eGFR and UACR to estimate the 2- and 5-year risk of kidney failure (International calculator: https://kidneyfailurerisk.co. uk).⁶³⁻⁶⁵ In the UK, KFRE >5% risk at 5 years should be used to guide patients who should be referred to nephrology.^{6,22}

The limitations of KFRE are that: (a) it provides a point estimate and does not account for variation in eGFR or UACR over time; (b) it does not account for the cause of kidney disease; and (c) it does not account for the fact that some patients may have a high KFRE estimate, but the competing event of death would probably occur before kidney failure. It is therefore appropriate to consider the KFRE result in the context of the individual patient: referral to nephrology is not mandated and may be inappropriate in patients with shorter life expectancy or in those who would not want or benefit from kidney replacement therapy.

WILEY 11

Recent work in the UK suggests that these criteria changes should not dramatically increase or decrease the number of referrals sent to nephrology from primary care.²² However, current low rates of albuminuria testing preclude the calculation of KFRE in about onethird of the CKD population; the desired improvements in albuminuria testing will probably identify greater numbers of patients meeting criteria for referral to nephrology.²²

3.8 | MANAGING CHRONIC KIDNEY DISEASE IN THE OLDER PATIENT

Kidney function decline is observed with increasing age, even among apparently healthy individuals; CKD, by eGFR criteria, is therefore likely to be observed in a substantial proportion of older individuals.^{66,67} Most recommendations for the investigation, diagnosis and management of CKD were based on general adult populations and may not be directly transferable to older adults, particularly those living with complex, multimorbid disease and/or frailty.

Patients living with CKD who are younger will have a higher lifetime risk of major adverse cardiovascular and renal events because of a longer expected life span to accrue the risks. Even among older, apparently healthy individuals, mild CKD remains associated with adverse outcomes and warrants consideration of treatment.^{5,68}

It is reasonable and appropriate to adjust the investigation, referral and treatment intensity according to the expected gain for an individual patient. Patients with the highest absolute level of risk, including older patients, those with multimorbidity and frailty, may expect to derive a higher absolute benefit from risk reduction strategies (as has been seen for blood pressure control and SGLT2i).^{69–72} However, there is also a higher rate of treatment-associated adverse events in these groups, and there is only sparse evidence to guide the risk/benefit balance.^{69–72} Minimizing medicalization and polypharmacy may be appropriate for those with a shorter life expectancy or to align with individuals' perspectives and treatment goals. The guidelines cannot cover all eventualities but allow the health care practitioner to exercise their clinical judgement in advising on management for an individual patient.

3.9 | CONCLUSION

CKD represents an increasingly common disease with major health and economic implications, but detection and awareness among health care providers and patients remain suboptimal. Early diagnosis of CKD, testing simple blood and urine markers in high-risk populations, is key to mitigating the risks associated with CKD. Timely risk-reduction strategies have overlapping indications to reduce kidney disease progression, cardiovascular disease and premature death. The pillars of CKD management in the community include lifestyle management, blood pressure and glycaemic control, statins and targeted interventions to reduce cardiorenal risk: RAASi, SGLT2i, GLP-1RA and nsMRA. New tools provide individualized risk assessment to guide the need for specialist care among patients at the highest level of risk. Although CKD is associated with higher risks of adverse outcomes even among older individuals, it is appropriate to individualize the approach for older individuals and among those who may not benefit from intensive investigation and treatment.

ACKNOWLEDGMENTS

This article was commissioned by the Editor as part of a Special Supplement made possible by an educational grant from Bayer Pharmaceuticals. Sponsor identity was not disclosed to the authors prior to publication.

CONFLICT OF INTEREST STATEMENT

Outside the submitted work, JSL has received personal lectureship honoraria from Astra Zeneca, Pfizer and Bristol Myers Squibb. The other authors have no conflicts of interest to declare.

PEER REVIEW

The peer review history for this article is available at https://www. webofscience.com/api/gateway/wos/peer-review/10.1111/dom.15625.

DATA AVAILABILITY STATEMENT

Data sharing is not applicable to this article as no new data were created or analyzed in this study.

ORCID

Jennifer S. Lees D https://orcid.org/0000-0001-6331-0178

REFERENCES

- Bello AK, Okpechi IG, Levin A, et al. An update on the global disparities in kidney disease burden and care across world countries and regions. *Lancet Global Health*. 2024;12:e382-e395. doi:10.1016/ S2214-109X(23)00570-3
- Kerr M, Bray B, Medcalf J, et al. Estimating the financial cost of chronic kidney disease to the NHS in England, *Nephrology Dialysis Transplantation*, 2012;27(Suppl 3):iii73-iii80. doi:10.1093/ndt/ gfs269
- Sullivan MK, Jani BD, McConnachie A, et al. Hospitalisation events in people with chronic kidney disease as a component of multimorbidity: parallel cohort studies in research and routine care settings. *BMC Med.* 2021;19:278. doi:10.1186/s12916-021-02147-6
- Stevens PE, Ahmed SB, Carrero JJ, et al. KDIGO 2024 clinical practice guideline for the evaluation and Management of Chronic Kidney Disease. *Kidney Int.* 2024;105:S117-S314. doi:10.1016/j.kint.2023. 10.018
- Writing Group for the CKD Prognosis Consortium, Grams ME, Coresh J, et al. Estimated glomerular filtration rate, albuminuria, and adverse outcomes: an individual-participant data meta-analysis. JAMA. 2023;330:1266-1277. doi:10.1001/jama.2023.17002
- National Institute for Health and Care Excellence. Chronic Kidney Disease in Adults: Assessment and Management [NG203]. NICE Guideline; 2021.
- Komenda P, Ferguson TW, Macdonald K, et al. Cost-effectiveness of primary screening for CKD: a systematic review. *Am J Kidney Dis*. 2014;63:789-797. doi:10.1053/j.ajkd.2013.12.012
- Levey AAS, Stevens LA LA, Schmid CH, et al. A new equation to estimate glomerular filtration rate. Ann Intern Med. 2009;150:604-612. doi:10.7326/0003-4819-150-9-200 905 050-00006
- Inker L, Eneanya N, Coresh J, et al. New creatinine- and cystatin Cbased equations to estimate GFR without race. N Engl J Med. 2021; 385:1737-1749. doi:10.1056/NEJMoa2102953

¹² WILEY-

- Hsu C, Yang W, Parikh RV, et al. Race, genetic ancestry, and estimating kidney function in CKD. N Engl J Med. 2021;385:1750-1760. doi: 10.1056/nejmoa2103753
- Pottel H, Björk J, Rule AD, et al. Cystatin C-based equation to estimate GFR without the inclusion of race and sex. N Engl J Med. 2023; 388:333-343. doi:10.1056/NEJMoa2203769
- 12. Perazella MA, Rosner MH. Drug-induced acute kidney injury. *Clin J Am Soc Nephrol*. 2022;17:1220-1233. doi:10.2215/CJN.11290821
- Carrero J-J, Fu EL, Sang Y, et al. Discordances between creatinineand cystatin C-based estimated GFR and adverse clinical outcomes in routine clinical practice. *Am J Kidney Dis.* 2023;82:534-542. doi:10. 1053/j.ajkd.2023.04.002
- Hanna PE, Wang Q, Strohbehn IA, et al. Medication-related adverse events and Discordancies in cystatin C-based vs serum creatininebased estimated glomerular filtration rate in patients with cancer. JAMA Netw Open. 2023;6:e2321715. doi:10.1001/jamanetworkopen. 2023.21715
- Lees JS, Fabian J, Shlipak MG. Cystatin C should be routinely available for estimating kidney function. *Curr Opin Nephrol Hypertens*. 2024;33:337-343. doi:10.1097/MNH.00000000000980
- Lees JS, Ho F, Parra-Soto S, et al. Kidney function and cancer risk: an analysis using creatinine and cystatin C in a cohort study. *EClinicalMedicine*. 2021;38:101030. doi:10.1016/j.eclinm.2021.101030
- Mok Y, Ballew SH, Sang Y, et al. Albuminuria, kidney function, and cancer risk in the community. *Am J Epidemiol*. 2020;189:942-950. doi: 10.1093/aje/kwaa043
- Coresh J, Heerspink HJL, Sang Y, et al. Change in albuminuria and subsequent risk of end-stage kidney disease: an individual participant-level consortium meta-analysis of observational studies. *Lancet Diabetes Endocrinol.* 2019;7:115-127. doi:10.1016/S2213-8587(18)30313-9
- Agarwal R, Tu W, Farjat AE, et al. Impact of Finerenone-induced albuminuria reduction on chronic kidney disease outcomes in type 2 diabetes: a mediation analysis. Ann Intern Med. 2023;176:1606-1616. doi:10.7326/M23-1023
- Oshima M, Neuen BL, Li J, et al. Early change in albuminuria with Canagliflozin predicts kidney and cardiovascular outcomes: a PostHoc analysis from the CREDENCE trial. J Am Soc Nephrol. 2020;31:2925-2936. doi:10.1681/ASN.2020050723
- de Boer IH, Khunti K, Sadusky T, et al. Diabetes Management in Chronic Kidney Disease: a consensus report by the American Diabetes Association (ADA) and kidney disease: improving global outcomes (KDIGO). *Diabetes Care*. 2022;45:3075-3090. doi:10.2337/dci22-0027
- Sullivan MK, Jani BD, Rutherford E, et al. Potential impact of NICE guidelines on referrals from primary care to nephrology: a primary care database and prospective research study. Br J Gen Pract. 2023; 73:e141-e147. doi:10.3399/BJGP.2022.0145
- Nitsch D, Caplin B, Hull S, Wheeler D, On Behalf of the National CKD Audit and Quality Improvement Programme in Primary Care. First National CKD Audit Report. Healthcare Quality Improvement Partnership; 2017.
- Keong F, Gander J, Wilson D, Durthaler J, Pimentel B, Barzilay JI. Albuminuria screening in people with type 2 diabetes in a managed care organization. *AJPM Focus*. 2023;2:100133. doi:10.1016/j.focus. 2023.100133
- 25. National Institute for Health and Care Excellence. *Hypertension in Adults: Diagnosis and Management [NG136] [Internet]*. National Institute for Health and Care Excellence; 2019.
- Unger T, Borghi C, Charchar F, et al. 2020 International Society of Hypertension Global Hypertension Practice Guidelines. *Hypertension*. 2020;75:1334-1357. doi:10.1161/HYPERTENSIONAHA.120. 15026
- Williams B, Mancia G, Spiering W, et al. 2018 ESC/ESH guidelines for the management of arterial hypertension. J Hypertens. 2018;36:1953-2041. doi:10.1097/HJH.00000000001940

- Sumida K, Nadkarni GN, Grams ME, et al. Conversion of urine protein-creatinine ratio or urine dipstick protein to urine albumincreatinine ratio for use in chronic kidney disease screening and prognosis. Ann Intern Med. 2020;173:426-435. doi:10.7326/M20-0529
- Kidney Disease: Improving Global Outcomes (KDIGO) Blood Pressure Work Group. KDIGO. Clinical practice guideline for the Management of Blood Pressure in chronic kidney disease. *Kidney International*. 2021;2021(99):S1-S87. doi:10.1016/j.kint.2020.11.003
- Whelton PK, Carey RM, Aronow WS, et al. 2017 ACC/-AHA/AAPA/ABC/ACPM/AGS/APhA/ASH/ASPC/NMA/PCNA guideline for the prevention, detection, evaluation, and Management of High Blood Pressure in adults: a report of the American College of Cardiology/American Heart Association task force on clinical practice guidelines. *Hypertension*. 2018;71(6):e136-e139. doi:10.1161/HYP. 000000000000065
- Mancia G, Kreutz R, Brunström M, et al. 2023 ESH guidelines for the management of arterial hypertension the task force for the management of arterial hypertension of the European Society of Hypertension. *J Hypertens*. 2023;41:1874-2071. doi:10.1097/HJH.00000000003480
- 32. National Institute for Health and Care Excellence. *Suspected Cancer: Recognition and Referral* [NG12]. NICC; 2015.
- Molokhia M, Okoli GN, Redmond P, et al. Uncoded chronic kidney disease in primary care: a cross-sectional study of inequalities and cardiovascular disease risk management. Br J Gen Pract. 2020;70: e785-e792. doi:10.3399/bjgp20X713105
- Bosi A, Xu Y, Gasparini A, et al. Use of nephrotoxic medications in adults with chronic kidney disease in Swedish and US routine care. *Clin Kidney J.* 2022;15:442-451. doi:10.1093/ckj/sfab210
- Swartling O, Yang Y, Clase CM, et al. Sex differences in the recognition, monitoring, and management of CKD in health care: an observational cohort study. J Am Soc Nephrol. 2022;33:1903-1914. doi:10. 1681/ASN.2022030373
- Hull SA, Rajabzadeh V, Thomas N, et al. Improving coding and primary care management for patients with chronic kidney disease: an observational controlled study in East London. Br J Gen Pract. 2019;69: e454-e461. doi:10.3399/bjgp19X704105
- Wong K, Pitcher D, Braddon F, et al. Effects of rare kidney diseases on kidney failure: a longitudinal analysis of the UK National Registry of rare kidney diseases (RaDaR) cohort. *Lancet*. 2024;403:1279-1289. doi:10.1016/S0140-6736(23)02843-X
- Chu CD, McCulloch CE, Banerjee T, et al. CKD awareness among US adults by future risk of kidney failure. *Am J Kidney Dis*. 2020;76:174-183. doi:10.1053/j.ajkd.2020.01.007
- Tuot DS, Zhu Y, Velasquez A, et al. Variation in Patients' awareness of CKD according to how they are asked. *Clin J Am Soc Nephrol*. 2016;11:1566-1573. doi:10.2215/CJN.00490116
- 40. Kidney Care UK. *Chronic Kidney Disease (CKD) [Internet*]. Kidney Care UK; [cited 2024 Mar 19].
- 41. Health Improvement Scotland. Scottish Intercollegiate Guidelines Network 149 Risk estimation and the prevention of cardiovascular disease [Internet]. Health Improvement Scotland; National Institute for Health and Care Excellence; 2017 [cited 2024 Mar 15].
- 42. Nuffield Department of Population Health Renal Studies Group. SGLT2 inhibitor Meta-Analysis Cardio-Renal Trialists' Consortium. Impact of diabetes on the effects of sodium glucose co-transporter-2 inhibitors on kidney outcomes: collaborative meta-analysis of large placebo-controlled trials. *Lancet.* 2022;400:1788-1801. doi:10.1016/ S0140-6736(22)02074-8
- Herrington WG, Emberson J, Mihaylova B, et al. Impact of renal function on the effects of LDL cholesterol lowering with statin-based regimens: a meta-analysis of individual participant data from 28 randomised trials. *Lancet Diabetes Endocrinol.* 2016;4:829-839. doi:10.1016/S2213-8587(16)30156-5
- 44. Xie X, Liu Y, Perkovic V, et al. Renin-angiotensin system inhibitors and kidney and cardiovascular outcomes in patients with CKD: a

Bayesian network meta-analysis of randomized clinical trials. *Am J Kidney Dis.* 2016;67:728-741. doi:10.1053/j.ajkd.2015.10.011

- The EMPA-KIDNEY Collaborative Group. Empagliflozin in patients with chronic kidney disease. N Engl J Med. 2023;388:117-127. doi:10. 1056/NEJMoa2204233
- UK Kidney Association, Herrington WG, Frankel AH. Sodium-glucose Co-transporter-2 (SGLT-2) inhibition in adults with kidney disease. 2023.
- Heerspink H, Stefansson B, Correa-Rotter R, et al. Dapagliflozin in patients with chronic kidney disease. N Engl J Med. 2021;384:388-390. doi:10.1056/nejmc2032809
- Judge P, Staplin N, Mayne K, et al. Impact of primary kidney disease on the effects of empagliflozin in patients with chronic kidney disease: secondary analyses of the EMPA-KIDNEY trial. *Lancet Diabetes Endocrinol.* 2024;12:51-60. doi:10.1016/S2213-8587(23)00322-4
- Heerspink HJL, Jongs N, Chertow GM, et al. Effect of dapagliflozin on the rate of decline in kidney function in patients with chronic kidney disease with and without type 2 diabetes: a prespecified analysis from the DAPA-CKD trial. *Lancet Diabetes Endocrinol.* 2021;9:743-754. doi:10.1016/S2213-8587(21)00242-4
- Wheeler DC, Toto RD, Stefánsson BV, et al. A pre-specified analysis of the DAPA-CKD trial demonstrates the effects of dapagliflozin on major adverse kidney events in patients with IgA nephropathy. *Kidney Int.* 2021;100:215-224. doi:10.1016/j.kint.2021.03.033
- Wheeler DC, Stefánsson BV, Jongs N, et al. Effects of dapagliflozin on major adverse kidney and cardiovascular events in patients with diabetic and non-diabetic chronic kidney disease: a prespecified analysis from the DAPA-CKD trial. *Lancet Diabetes Endocrinol*. 2021;9:22-31. doi:10.1016/S2213-8587(20)30369-7
- Cherney DZI, Cooper ME, Tikkanen I, et al. Pooled analysis of phase III trials indicate contrasting influences of renal function on blood pressure, body weight, and HbA1c reductions with empagliflozin. *Kidney Int.* 2018;93:231-244. doi:10.1016/j.kint.2017.06.017
- Bakris GL, Agarwal R, Anker SD, et al. Effect of Finerenone on chronic kidney disease outcomes in type 2 diabetes. N Engl J Med. 2020;383: 2219-2229. doi:10.1056/NEJMoa2025845
- Pitt B, Filippatos G, Agarwal R, et al. Cardiovascular events with Finerenone in kidney disease and type 2 diabetes. N Engl J Med. 2021;385:2252-2263. doi:10.1056/NEJMoa2110956
- Kidney Disease: Improving Global Outcomes (KDIGO) Diabetes Work Group. KDIGO. Clinical practice guideline for diabetes Management in Chronic Kidney Disease. *Kidney Int.* 2022;2022(102):S1-S127. doi: 10.1016/j.kint.2022.06.008
- National Institute for Health and Care Excellence. Finerenone for Treating Chronic Kidney Disease in Type 2 Diabetes [TA877]. National Institute for Health and Care Excellence; 2023.
- Department of Health and Social Care NE. Shortage of GLP-1 Receptor Agonists (GLP-1 RA) Update. Department of Health and Social Care, NHS England; 2024.
- Baker M, Perazella MA. NSAIDs in CKD: are they safe? Am J Kidney Dis. 2020;76:546-557. doi:10.1053/j.ajkd.2020.03.023
- UK Health Security Agency, Department of Health and Social Care. Immunisation of Individuals with Underlying Medical Conditions: the Green Book, Chapter 7. Department of Health and Social Care, NHS England; 2020.

- 60. Bell S, Campbell J, McDonald J, et al. COVID-19 in patients undergoing chronic kidney replacement therapy and kidney transplant recipients in Scotland: findings and experience from the Scottish renal registry. *BMC Nephrol.* 2020;21:419. doi:10.1186/s12882-020-02061-8
- Babel N, Hugo C, Westhoff TH. Vaccination in patients with kidney failure: lessons from COVID-19. *Nat Rev Nephrol.* 2022;18:708-723. doi:10.1038/s41581-022-00617-5
- Sullivan MK, Lees JS, Drake TM, et al. Acute kidney injury in patients hospitalized with COVID-19 from the ISARIC WHO CCP-UK study: a prospective, multicentre cohort study. *Nephrol, Dial, Transplant.* 2022; 37:271-284. doi:10.1093/ndt/gfab303
- Tangri N, Stevens LA, Griffith J, et al. A predictive model for progression of chronic kidney disease to kidney failure. JAMA. 2011;305(15). doi:10.1001/jama.2011.451
- Tangri N, Grams ME, Levey AS et al. Multinational assessment of accuracy of equations for predicting risk of kidney failure: a metaanalysis. JAMA. 2016;315(2):1-11.
- 65. Major RW, Shepherd D, Medcalf JF, Xu G, Gray LJ, Brunskill NJ. The kidney failure risk equation for prediction of end stage renal disease in UK primary care: an external validation and clinical impact projection cohort study. *PLoS Medicine*. 2019;16(11):e1002955.
- Eriksen BO, Palsson R, Ebert N, et al. GFR in healthy aging: an individual participant data meta-analysis of iohexol clearance in european population-based cohorts. J Am Soc Nephrol. 2020;31:1602-1615. doi:10.1681/ASN.2020020151
- Melsom T, Norvik JV, Enoksen IT, et al. Sex differences in age-related loss of kidney function. J Am Soc Nephrol. 2022;33:1891-1902. doi: 10.1681/asn.2022030323
- Lees JS, Rutherford E, Stevens KI, et al. Assessment of cystatin C level for risk stratification in adults with chronic kidney disease. JAMA Netw Open. 2022;5:e2238300. doi:10.1001/jamanetworkopen.2022. 38300
- Bress AP, Greene T, Derington CG, et al. Patient selection for intensive blood pressure management based on benefit and adverse events. J Am Coll Cardiol. 2021;77:1977-1990. doi:10.1016/j.jacc. 2021.02.058
- Vart P, Butt JH, Jongs N, et al. Efficacy and safety of Dapagliflozin in patients with chronic kidney disease across the Spectrum of frailty. *J Gerontol, Ser A.* 2024;79. doi:10.1093/gerona/glad181
- Butt JH, Dewan P, Merkely B, et al. Efficacy and safety of Dapagliflozin according to frailty in heart failure with reduced ejection fraction. *Ann Intern Med.* 2022;175:820-830. doi:10.7326/M21-4776
- Butt JH, Jhund PS, Belohlávek J, et al. Efficacy and safety of Dapagliflozin according to frailty in patients with heart failure: a Prespecified analysis of the DELIVER trial. *Circulation*. 2022;146:1210-1224. doi: 10.1161/CIRCULATIONAHA.122.061754

How to cite this article: Mayne KJ, Hanlon P, Lees JS. Detecting and managing the patient with chronic kidney disease in primary care: A review of the latest guidelines. *Diabetes Obes Metab.* 2024;1-13. doi:10.1111/dom.15625